2014s-47

Mandatory minimum contributions, heterogeneous endowments and voluntary public-good provision

Claudia Keser, Andreas Markstädter, Martin Schmidt

Série Scientifique Scientific Series

Montréal Décembre/December 2014

© 2014 Claudia Keser, Andreas Markstädter, Martin Schmidt. Tous droits réservés. All rights reserved. Reproduction partielle permise avec citation du document source, incluant la notice ©. Short sections may be quoted without explicit permission, if full credit, including © notice, is given to the source.



Centre interuniversitaire de recherche en analyse des organisations

CIRANO

Le CIRANO est un organisme sans but lucratif constitué en vertu de la Loi des compagnies du Québec. Le financement de son infrastructure et de ses activités de recherche provient des cotisations de ses organisations-membres, d'une subvention d'infrastructure du Ministère de l'Économie, de l'Innovation et des Exportations, de même que des subventions et mandats obtenus par ses équipes de recherche.

CIRANO is a private non-profit organization incorporated under the Québec Companies Act. Its infrastructure and research activities are funded through fees paid by member organizations, an infrastructure grant from the Ministère de l' l'Économie, de l'Innovation et des Exportations, and grants and research mandates obtained by its research teams.

Les partenaires du CIRANO

Partenaire majeur Ministère de l'Économie, de l'Innovation et des Exportations

Partenaires corporatifs

Autorité des marchés financiers Banque de développement du Canada Banque du Canada Banque Laurentienne du Canada Banque Nationale du Canada Bell Canada BMO Groupe financier Caisse de dépôt et placement du Québec Fédération des caisses Desjardins du Québec Financière Sun Life, Québec Gaz Métro Hydro-Québec Industrie Canada Intact Investissements PSP Ministère des Finances et de l'Économie Power Corporation du Canada Rio Tinto Alcan Ville de Montréal

Partenaires universitaires

École Polytechnique de Montréal École de technologie supérieure (ÉTS) HEC Montréal Institut national de la recherche scientifique (INRS) McGill University Université Concordia Université de Montréal Université de Sherbrooke Université du Québec Université du Québec Université du Québec à Montréal Université Laval

Le CIRANO collabore avec de nombreux centres et chaires de recherche universitaires dont on peut consulter la liste sur son site web.

Les cahiers de la série scientifique (CS) visent à rendre accessibles des résultats de recherche effectuée au CIRANO afin de susciter échanges et commentaires. Ces cahiers sont écrits dans le style des publications scientifiques. Les idées et les opinions émises sont sous l'unique responsabilité des auteurs et ne représentent pas nécessairement les positions du CIRANO ou de ses partenaires.

This paper presents research carried out at CIRANO and aims at encouraging discussion and comment. The observations and viewpoints expressed are the sole responsibility of the authors. They do not necessarily represent positions of CIRANO or its partners.

ISSN 2292-0838 (en ligne)

Partenaire financier Économie, Innovation et Exportations Québec 😫 🏟

Mandatory minimum contributions, heterogeneous endowments and voluntary public-good provision

Claudia Keser^{*}, Andreas Markstädter[†], Martin Schmidt[‡]

Résumé/abstract

In a public-good experiment with heterogeneous endowments, we investigate if and how the contribution level as well as the previously observed "fair-share" rule of equal contributions relative to one's endowment (Hofmeyr et al., 2007; Keser et al., 2014) may be influenced by minimum-contribution requirements. We consider three different schedules: FixMin, requiring the same absolute contributions, RelMin, requiring the same relative contributions, and ProgMin, requiring minimum contributions that progressively increase with the endowment. We find that minimum contributions exert norm-giving character and may lead to an increase in average group contributions. This is especially true for the progressive schedule. On the individual level, this schedule leads to higher relative contributions by the wealthier players and thus violates the "fair-share" norm. On the group level, it leads to the highest contribution level and the lowest inequality in total profits as measured by the Gini index.

Mots clés /Key words : Experimental economics, public goods, heterogeneous endowments, mandatory minimum contributions, norms.

Codes JEL : C92, D63, H41

[†] Chair of Microeconomics, Faculty of Economic Sciences, Georg-August-Universität Göttingen, Platz der

^{*} *Corresponding Author*: Chair of Microeconomics, Faculty of Economic Sciences, Georg-August-Universität Göttingen, Platz der Göttinger Sieben 3, 37073 Göttingen, Germany, Tel.: +49 (0) 551 - 39 7336, <u>claudia.keser@uni-goettingen.de</u>. Fellow CIRANO, Montréal, Québec, Canada.

Göttinger Sieben 3, 37073 Göttingen, Germany, Tel.: +49 (0) 551 - 39 7336, andreas.markstaedter@uni-goettingen.de

[‡] Chair of Microeconomics, Faculty of Economic Sciences, Georg-August-Universität Göttingen, Platz der Göttinger Sieben 3, 37073 Göttingen, Germany, Tel.: +49 (0) 551 - 39 7336, martin.schmidt@wiwi.uni-goettingen.de

1. Introduction

When it comes to their funding, several public institutions, such as, for example, museums, theaters, and operas, rely on a two-tier model. They apply mandatory admission charges that may be voluntarily supplemented by charitable donations. Given that these institutions provide merit goods, their two-tier funding situation may be modeled in a public-good game, where mandatory levies are requested and additional contributions are possible. Such a model has been introduced by James Andreoni (1993) to investigate crowding-out effects of public intervention in a laboratory experiment. He observes that voluntary contributions are partially crowded out by a lump-sum "tax". It still remains an open question, though, how such mandatory levies should be designed to maximize the revenues of public institutions, taking into account that agents possess unequal economic possibilities (wealth) and differ in their willingness to pay.

An extensive experimental literature on the voluntary contribution mechanism (VCM) finds that behavior in public-good games cannot solely be explained by standard economic preferences. Contributions, though declining over time, are generally higher than the Nash-equilibrium prediction. This is true whether the dominant strategy is to contribute nothing (e.g., Marwell and Ames, 1979, 1980) or whether it lies in the interior of the strategy space (e.g., Keser, 1996). Survey studies suggest that, when the dominant strategy is to contribute nothing and participants are equally endowed, initial contributions typically lie between 40 and 60 percent of the endowment but decay over time. In the last round about 70 percent of subjects contribute nothing (e.g., Davis and Holt, 1993; Dawes and Thaler, 1988; Ledyard, 1995; Ostrom, 2000). Although contributions are higher than theoretically predicted, they are at a considerable distance from the social optimum. There is ample evidence that subjects tend to coordinate their contributions by conditional cooperation, i.e., they begin cooperatively and reciprocate depending on the actions of others (e.g., Keser and van Winden, 2000; Fischbacher et al., 2001). How such cooperation is affected by an inequality in endowments (wealth) is, however, not yet well understood. John Ledyard's conjecture that homogeneity in endowments has a positive effect on group contributions, or in other words, that heterogeneity has negative effects (Ledyard, 1995), has only partly been confirmed in the recent literature. Group contributions in weakly asymmetric environments tend to be equal to those in symmetric settings, where the same total endowment is allocated evenly over all group members. Thereby, players tend to follow a "fair-share" rule, where they coordinate on equal relative contributions of the endowment (Hofmeyr et al., 2007; Keser et al., 2014; Keser and Schmidt, 2014). The "fair-share" rule has its limitations, though. When the asymmetry in the endowments becomes so large that one of the players loses interest in the group optimum, the norm shifts from equal relative to equal absolute contributions and the group contribution level declines significantly (Keser et al., 2014).

In this study, we consider an environment with heterogeneous endowments where the "fair-share" norm applies (the AsymWeak treatment of Keser et al. (2014)) and investigate if and how this norm as well as the group contribution level may be influenced by minimum contribution requirements similar to the "taxes" in Andreoni (1993). In our experiment, we implement various minimum-contribution schedules (MCS) charging subjects with different endowments with different levies. These levies are compulsory minimum contributions. The staggering of our MCS is motivated by common tax structures to be found in real-life settings. We consider a treatment with a lump-sum "tax" (FixMin), requiring the same absolute minimum contributions from all players, a treatment with a flat "tax" rate (RelMin) that requires the same minimum contribution relative to their respective levels of endowment from all players, and a treatment with a progressive "tax" schedule (ProgMin), where the more wealthy players are requested to provide a higher minimum contribution relative to their endowment than the less wealthy ones. The AsymWeak treatment of Keser et al. (2014) serves as the baseline treatment (NoMin) without any minimum-contribution requirement.¹ We do not see the mandatory levies in our study as taxes in the strict sense. Tax burdens impose tax levels, which are not intended to be overspent by taxpayers. Our minimumcontribution schedules more closely resemble the mandatory admission charges mentioned above. We consider them as a policy tool to potentially impose norms.

Despite the fact that in our experimental game there are no "standard economic" incentives for contributions above the compulsory minimum values, we expect our different minimum contribution schemes to exert "expressive power" (e.g., Cooter, 1998; Galbiati and Vertova, 2008), while driving contributions beyond the enforced minima due to incomplete crowding-out (Andreoni, 1993). Thereby, we anticipate that the miscellaneous schedules differently affect the sense of justice and the willingness to contribute among the group members. To investigate this, we define a measure of motivational crowding-out for our experimental setup. This measure relates to the classic crowding-out in the analysis of public policy (Andreoni, 1993; Andreoni and Payne, 2011) but it is different in that it considers the percentage of the freely disposable endowment that is contributed, rather than the absolute amount contributed. The motivational crowding-out measure indicates zero crowding-out if subjects that contribute, for example, 50 percent of their endowment in the absence of mandatory minimum contribution requirements will still contribute 50 percent of their freely disposable endowment (endowment beyond the mandatory contribution) in the presence of such requirements. Full motivational crowding-out implies contributions equal to the mandatory contribution levels.

In the choice of the different minimum-contribution schedules, we pay attention to two features. Firstly, the total amount of minimum contributions is constant across all treatments. Secondly, minimum contributions for the individual player types are lower

 $^{^{1}}$ The differentiation of mandatory contribution levels in the treatments with the relative and the progressive MCS can be seen as a kind of third degree price discrimination, where the levies vary by wealth status.

than the respective average contributions in the baseline treatment (without mandatory minimum contributions) and thus sum up to less than the (unenforced) average total group contribution in the baseline treatment.² This is necessary, since we are not interested in increasing contributions to the public good by mandatory contributions that are high enough to exceed the amount that people would contribute voluntarily, anyway. We strive to investigate if and how different distributions of a given total mandatory levy may change individual and group contribution patterns.

As discussed above, our study relates to two literature strands. The first one deals with the "expressive power" of law and is thus particularly relevant to the part of our study that deals with the impact of the various minimum contribution schemes on contribution norms. Law can be defined as an obligation, and, according to the expressive-power hypothesis, it might have psychological effects on individual preferences. In other words, actors might view an obligation as an internal value. Law can thus create a focal point by creating values (e.g., Cooter, 1998; Galbiati and Vertova, 2008). The second literature strand deals with the impact of external interventions on intrinsic motivation with respect to crowding-out or crowding-in (e.g., Deci et al., 1999). Given the finding of incomplete crowding-out of voluntary contributions by minimum-contribution requirements (Andreoni, 1993; Chan et al., 2002), we expect our tax systems to increase group contributions. Both literature strands are presented in more detail in Section 2.

The main findings of our experiments can be summarized as follows. We find that mandatory minimum contributions do have a norm-giving character (expressive power). Group contribution levels are significantly higher in ProgMin than in NoMin and FixMin. On the individual level, we observe in all treatments that individuals with higher endowments make on average higher contributions than those with lower endowments. Furthermore, in RelMin, we replicate the "fair-share" rule observed in NoMin (Keser et al., 2014). In FixMin, we find relative contributions to be higher for the less than for the more wealthy players, but we observe the opposite in ProgMin. This suggests that the contribution norm can indeed be influenced through a deliberate intervention like a minimum contribution requirement. As a consequence, the progressive contribution schedule leads to a significant increase in average group contributions relative to the baseline treatment without minimum contribution requirements. While we observe motivational crowding-out in FixMin, we have no evidence for motivational crowding-out in ProgMin and RelMin.

The remainder of this paper is structured as follows. Section 2 gives a short overview over the related literature. Section 3 presents the experimental design and derives

² To investigate crowding-out, Andreoni (1993) chooses a "tax" below the interior solution of the nonlinear public-good game. Our linear public good game does not have an interior Nash equilibrium but has a dominant strategy solution to contribute nothing. The outcome of the baseline treatment, however, lies in the interior of the strategy space and can be presented as a quantal response equilibrium under the assumption of altruism and error making (Anderson et al., 1998).

testable hypotheses. Section 4 reviews these hypotheses in the face of the experimental results. Section 5 provides a summary and conclusions.

2. Related literature

We consider two ways how mandatory minimum contributions schedules could impact voluntary contributions to a public good. The first is derived from the literature on expressive law that hypothesizes that obligations have a potential to influence behavior. They may create focal points or norms, which channel individuals' beliefs about the behavior of others and act as coordination devices (Cooter, 1998; Galbiati and Vertova, 2008; McAdams and Nadler, 2005). Rational individuals internalize a norm (i.e., change their behavior) when commitment promises an advantage (Cooter, 1998). A norm set by the mandatory minimum contribution levels that is perceived as appropriate to enhance one's profit is hence potentially able to increase individual and group contributions.

Galbiati and Vertova (2008), for example, study expressive law with weakly incentivized non-binding obligations in a public-good game. These obligations are presented as minimum contributions that are not mandatory and thus leave the players' decision spaces unaffected. However, participants know that they will be probabilistically audited and penalized or rewarded if they have under- or over-fulfilled their obligations. In a repeated linear public-good game with groups of six equally endowed subjects, Galbiati and Vertova test whether different obligation levels imply different levels of cooperation. They find that obligations in repeated interactions significantly affect the average level of individual contributions and the rate of decrease in cooperation over time. Higher obligations reduce the pace of the decline in average contributions. Unexpected changes in the level of minimum contributions have asymmetric effects on the level of cooperation: "a reduction does not alter the pattern of deterioration of cooperation over time, whereas an increase triggers a re-start in cooperation" (p. 148). In a follow-up study, Galbiati and Vertova (2014) disentangle the effects of obligations and incentives. They consider non-binding incentives (such that zero contribution to the public good still remains the dominant strategy for risk-neutral players) with a low and a high probability of an audit. They find obligations and non-binding incentives to be complementary, jointly supporting high levels of contribution. Incentives alone do not significantly increase contributions, while high obligations in the form of recommendations moderately increase them.

In a similar study, using a repeated public-good game with groups of two identically endowed subjects, Riedel and Schildberg-Hörisch (2013) find that obligations increase contributions in the first rounds. Contributions toward the end of the game, however, are not statistically distinguishable to the case without obligations. Individual contributions are affected by the own obligation but independently from the partners obligation. For a given obligation, behavior is not significantly different between symmetric and asymmetric obligation treatments. However, the fraction of noncompliers is higher for subjects with higher inflicted obligation. Given that the decline in contributions is only significant for individuals with high obligations, subjects seem to incur non-monetary costs while disobeying obligations. People are not more likely to violate an obligation if it is perceived unfair.

The three studies above differ from our study in that the therein employed obligations, i.e., minimum contributions, are not binding. That means that in these studies participants face the same decision space whether obligations are imposed or not, i.e., under- or over-fulfillment of the obligation is possible. In our study the decision space is reduced by the minimum contribution requirements in the MCS treatments such that only over-fulfillment of the obligation is possible.

The second potential impact channel of minimum contributions relates to the literature on the impact of external interventions on intrinsic motivation. According to Ryan and Deci (2000) "[i]ntrinsic motivation is defined as the doing of an activity for its inherent satisfaction rather than for some separable consequence. When intrinsically motivated a person is moved to act for the fun or challenge entailed rather than because of external prods, pressures, or rewards" (p. 56). It is argued that, given a task that is performed voluntarily or for the sake of its own, any form of outside interference may affect the intrinsic motivation on which the initial action is based and thus change the amount of effort exerted in the task. Crowding-out of intrinsic motivation is reinforced, when the external intervention is perceived as controlling (Falk and Kosfeld, 2006) or when it has a negative effect on people's feeling of self-determination, competence and self-esteem (Deci, 1971; Nyborg and Rege, 2003; Rotter, 1966).

Titmuss (1970) was the first to establish the hypothesis that monetary rewards may crowd out intrinsic motivation. He came up with the example of blood donations, where present donors may reduce donations, if they perceive that their intrinsic motivation is not appreciated, when monetary incentives for their donations are offered. The result is what Condry and Chambers (1978) call "hidden costs of reward" as rewards tend to distract attention from the process of the task activity itself to the goal of getting a reward. This hypothesis was confirmed by Uptom (1973). Since Titmuss (1970) a large body of literature found indication for his hypothesis in a variety of other circumstances³. Note, however, that in contrast to the said, interventions, might also be perceived as supportive and promote self-esteem. If this is the case, they might even crowd in intrinsic motivation (Frey and Jegen, 2001).

In the case of the funding of a public good, a specific additional factor could play a crucial role: the perception of one's moral responsibility for the provision of the good. If the government enforces minimum contribution levels, individuals may perceive a decay of

³ See, for example, Deci (1971, 1972), Deci et al. (1999), Frey (1993, 1994, 1997a, 1997b), Frey et al. (1996), Frey and Oberholzer-Gee (1997), Frey and Goette (1999), Frey and Jegen (2001), Gneezy and Rustichini (2000a, 2000b), and Lepper et al. (1973).

their responsibility for the provision, leading to a crowding-out of (additional) voluntary contributions. If however, the intervention is able to communicate morally ideal contribution levels, which are perceived as symbolic, even a crowding-in is possible (Brekke et al., 2003; Nyborg and Rege, 2003).

Theoretical models on government intervention in the provision of public goods, for models with an interior equilibrium, predict complete crowding-out of private voluntary by public contributions (e.g., Warr, 1982, 1983; Roberts, 1984, 1987; Bergstrom et al., 1986; Bernheim, 1986). Indeed, crowding-out of voluntary public-good provision by governmental provision is found in several empirical studies. In the context of charitable giving Abrams and Schmitz (1978, 1984) find crowding-out of about 30 percent and that in addition to governmental charitable payments the need of the recipients plays a crucial role in the decision for private charitable donations. Payne (1998) observes crowding-out of about 50 percent of private donations to non-profit organizations (NGOs) with increased government funding. In a laboratory experiment, based on a public-good game with an interior equilibrium, Andreoni (1993) finds crowding-out of about 70 percent by mandatory contributions. In a similar study Chan et al. (2002) also find that crowding-out is incomplete and that enforced contributions significantly increase total contributions to the public good.

3. The experiment

3.1. The game

In our public-good game *n* players form a group. Each player *i* (*i* = 1, ..., *n*) is endowed with a number of tokens, e_i , which have to be allocated between a private and a public investment. Let x_i denote the amount allocated to the private investment and y_i the amount allocated to the public investment by player *i*, with $x_i, y_i \ge 0$. The investments have to be in entire tokens and have to add up to the endowment. Furthermore, there is a minimum contribution requirement to the public investment c_i , with $c_i \ge 0$. Thus, $0 \le x_i \le e_i - c_i, c_i \le y_i \le e_i$, and $x_i + y_i = e_i$. The profit of each player *i* depends on his individual private investment and the sum of all public investments in his group. Each token that he allocates to the private investment yields him an individual return of α , while each token that he allocates to the public investment yields him and each other group member a return of β , with $\alpha > \beta$ and $n\beta > \alpha$. The profit function of player *i* can thus be written as:

$$\pi_i \left(x_i, \sum_{i=1}^n y_i \right) = \alpha x_i + \beta \sum_{i=1}^n y_i \tag{1}$$

Since a player's individual return in the private investment is larger than in the public investment ($\alpha > \beta$), the game has an equilibrium in dominant strategies, where each

player contributes the required minimum to the public investment and all remaining tokens to his private investment ($x_i^* = e_i - c_i$, $y_i^* = c_i$). If this game is played over a finite number of *T* rounds, the subgame perfect equilibrium solution prescribes, based on backward induction, that in each round $t \in (1, ..., T)$ each player contributes the required minimum to the public investment and all remaining tokens to his private investment ($x_{it}^* = e_i - c_i$, $y_{it}^* = c_i$).

Given that $n\beta > \alpha$, the sum of profits of all *n* players is maximized if all tokens are allocated to the public investment. Hence, in the social optimum all players allocate in each round their entire endowment to the public investment. The game-theoretical solution (subgame-perfect equilibrium) is thus collectively inefficient.

Given the evidence from earlier experiments on this kind of linear public-good game, where contributions, significantly deviate from the Nash equilibrium solution, also other equilibrium concepts are conceivable. For example, the so-called quantal response equilibrium, which is based on the assumption that subjects' decisions are determined by altruism and decision-error, can explain why mean contributions deviate from the Nash equilibrium (Anderson et al., 1998). Based on the quantal response equilibrium concept, the crowding-out measure by Andreoni (1993), which is based on the assumption that the Nash equilibrium (before and after taxation) lies in the interior of the decision space and that the lump-sum tax is smaller than the equilibrium contribution, can also be used in the context of our study to evaluate the impact of our minimum contribution schedules on individual and group contributions.

3.2. Procedure

We conducted the computerized experiment in the Göttingen Laboratory of Behavioral Economics at the University of Göttingen, Germany, based on the z-tree software package (Fischbacher, 2007).⁴ Participants were 160 bachelor and master students from various disciplines (mostly economics and business administration). Recruited via ORSEE (Greiner, 2004), they had previously volunteered to participate in decision-making experiments. On average, a roughly equal number of female and male students participated in the experiments; the number of women and men approximately balanced during all sessions. According to subject availability, we conducted sessions with three to four groups each, implying three to four independent observations per session. In total 40 independent observations were collected in four different treatments.

The procedure of the experiment was as follows. Upon arrival in the meeting room each participant got a randomly assigned participation number corresponding to a computer

⁴ The lab consists of 24 computers in isolated booths, such that vision of someone else's computer screen or verbal communication with other participants is highly restricted.

terminal in the lab. As soon as the required number of participants had shown up, the experimenter distributed written instructions (a translation of these is provided in Appendix B) and read them aloud to all participants.

Participants were informed that they would be randomly assigned to groups remaining unchanged during the entire experimental session (partners design). Participants, however, did not get to know the identity of the participants with whom they interacted. Each player in a group was randomly assigned a player number from one to four, which was individually communicated at the beginning of the experiment and remained unchanged. Each player number was associated with a certain fixed endowment and minimum contribution requirement per round. At the end of each round, participants were informed about the contributions to the public investment by each of the other players in the group (identified by their player number but otherwise anonymous), the total group contribution, the profit for the current round, and the total profit so far. Moreover, all participants were provided with a history of all previous rounds, containing the same information, on the screen.

After the reading of the instructions participants were seated at their respective computer terminals. Before the experiment started, we used computerized control questions with regard to the understanding of the instructions. The experiment did not start until all participants had provided correct answers to all questions.

The participants were informed in the instructions that the profits gained in the course of the experiment were measured in Experimental Currency Unit (ECU) and that these profits were to be multiplied by a conversion factor of $0.01 \in$ per ECU (which is the same for all players) for the final payment, in addition to a show-up fee of $3 \in$. The cash payment was conducted anonymously after the experiment.

An experimental session lasted on average around 75 minutes. The average payoff earned was about $15.50 \notin$ (including a $3 \notin$ show-up fee).

3.3. Parameters and treatments

Participants are assigned to groups of four (n = 4), the game is played for 25 rounds (T = 25), and the parameters of the profit function are $\alpha = 2$ and $\beta = 1$ (which implies a constant marginal per capita return (MPCR⁵) of 0.5 for the investment in the public account). Furthermore, participants are informed that Player 1 (Type 10) is endowed with 10 ECU, Players 2 and 3 (Type 15) are endowed with 15 ECU, each, and that Player 4 (Type 20) is endowed with 20 ECU. In each round, each participant has to make an allocation decision, conditioned on his minimum contribution requirement to

⁵ The MPCR is defined as the ratio of the private value of one token invested into the public investment to the private value of one token invested into the private investment.

the public good (c_i) . Minimum contribution requirements of all player types are common knowledge.

Table 1 presents the individual minimum contribution requirements (c_i) for the three player types in our four treatments: (1) NoMin, (2) FixMin, (3) RelMin, and (4) ProgMin. Under NoMin no participant is forced to contribute a mandatory levy; under FixMin every participant is forced to contribute a mandatory levy of 6 ECU; under RelMin each participant is forced to contribute a mandatory levy of 40 percent of her/his endowment; and under ProgMin each Type 10 player has to contribute 2 ECU, each Type 15 player has to contribute 6 ECU, and each Type 20 player has to contribute 10 ECU. The amounts of 2 ECU, 6 ECU, and 10 ECU in ProgMin correspond to progressive MCS rates of 20, 40, and 50 percent, for the three player types. The FixMin treatment is in principle a regressive MCS system, in that poorer individuals have to contribute relatively more of their endowment. The amounts of 6 ECU correspond to a regressive MCS regime with rates of 60, 40, and 30 percent for the three player types, respectively. In all treatments the total endowment of the four players is equal to 60 and in all MCS treatments the total mandatory levy is equal to 24.

		Min	imum con	tributions	$S(C_i)$		
Treatment	Lahel	Type	Ту 1	pe 5	Type 20	$\sum_{i=1}^{4} c_{i}$	# Obs
no.	Laber	Player	Player	Player	Player	$\sum_{i=1}^{c_i}$	# 003.
		1	2	3	4		
1	NoMin	0	0	0	0	0	10
		(0%)	(0%)	(0%)	(0%)	-	
2	FixMin	6	6	6	6	24	10
_		(60%)	(40%)	(40%)	(30%)		20
3	RelMin	4	6	6	8	24	10
5	Renvini	(40%)	(40%)	(40%)	(40%)	21	10
4	ProgMin	2	6	6	10	24	10
т	Tiogimin	(20%)	(40%)	(40%)	(50%)	24	10

Table 1:

Treatment overview

3.4. Hypotheses

To facilitate the illustration of the results in the following section our analysis focuses on two hypotheses.

Hypothesis 1: Minimum contribution requirements incompletely crowd out voluntary contributions implying that we observe higher group contributions in the MCS treatments than in NoMin.

Andreoni (1993) and Chan et al. (2002) measure crowding-out by $(\bar{Y}_0 + C - \bar{Y}_C)/C$, where \bar{Y}_0 is the average group contribution to the public good in NoMin, *C* is the sum of minimum contributions of all group members, and \bar{Y}_C is the average group contribution to the public good in the respective MCS treatment. Thus, crowding-out is 0 percent if $\bar{Y}_C = \bar{Y}_0 + C$ and it is 100 percent if $\bar{Y}_0 = \bar{Y}_C$. Based on this measure, they find that crowding-out is incomplete and, thus, that their public policy interventions by enforced minimum contributions significantly increase total contributions to the public good. Therefore, we expect that the three minimum contribution schedules increase group contribution requirements.

Hypothesis 2: Players follow a simple "fair-share" rule of equal relative contributions of the endowment in RelMin but not in FixMin and ProgMin.

Keser et al. (2014) have shown that players in NoMin tend to coordinate their contributions by using the simple "fair-share" rule, where they contribute equal amounts relative to the endowment. Since in RelMin mandatory contributions are staggered relative to the endowment, we expect contributions to follow this rule as well. However, we expect that Type 10 players contribute a higher (lower) share of their endowment than both other types in FixMin (ProgMin), and that Type 20 players contribute a higher (lower) share of their endowment than both other types in ProgMin (FixMin). In other words, we expect the proportional mandatory contributions in RelMin, the regressive mandatory contributions in FixMin, and the progressive mandatory contributions in ProgMin, respectively, to exert their "intended" influence by pushing individual contributions in the direction in which the minimum contribution requirements are staggered). We derive support for this conjecture from the literature on expressive law. This literature suggests that mandatory minimum contributions schedules may exert "expressive power" through the imposed obligations by expressing certain levels of "fair contribution" (Galbiati and Vertova, 2008; Riedel and Schildberg-Hörisch, 2013).

4. Results

Beyond the analysis of group contributions in Subsection 4.1 (Hypothesis 1) and individual contributions in subsection 4.2 (Hypothesis 2), we shall investigate reciprocity in Subsection 4.3 and profits in Subsection 4.4. All nonparametric tests presented in the following subsections are two-sided. We shall denote the Mann-

Whitney U test as U test and the Wilcoxon matched-pairs signed-rank test as signed-rank test.

4.1. Group contributions

Figure 1 presents the development of the average group contributions to the public investment (\bar{Y}_t) in the four treatments over the 25 rounds. As can be seen, the ProgMin treatment exhibits the highest contributions.⁶ Due to the large end-game effect in NoMin (in approximately the last five rounds), which is impeded by the minimum contribution requirements in the MCS treatments, we report only figures for rounds 1 to 20 in the forthcoming analysis. For this game interval, average group contributions are 36.4 in NoMin, 38.7 in FixMin, 40.2 in RelMin, and 44.7 in ProgMin.⁷ This indicates that the mandatory contributions only partially crowd out voluntary contributions. Following Andreoni (1993), crowding-out is 90.2 percent in FixMin, 83.8 percent in RelMin and as little as 65.0 percent in ProgMin. Pairwise treatment comparisons, based on U tests, show that average group contributions in ProgMin are significantly higher than those in NoMin (p = .0343) and FixMin (p = .0963); no other pairwise comparison shows a statistically significant difference.

It is important to mention that group contributions in ProgMin do not start from the beginning on a higher level than in the other two MCS treatments. They rather rise during the first rounds of the game and then remain nearly constant until the end-game. In the first round, we find that average group contributions do not differ between the MCS treatments, with respective figures of 39.0, 36.5, and 38.5 in FixMin, RelMin, and ProgMin. The average first-round contribution in NoMin is 30.8; it is significantly lower than in FixMin (p = .0884) and in ProgMin (p = .0487), when compared by U tests.⁸

Result 1: Group contribution levels are significantly higher in ProgMin than in NoMin and FixMin. For ProgMin, this confirms Hypothesis 1 that crowding-out is incomplete.

⁶ Over rounds 1-25, the mean group contributions are 43.5 in ProgMin compared to 33.1 in NoMin, 37.7 in FixMin, and 39.8 in RelMin. The respective average standard deviations are 7.5, 12.4, 5.9, and 6.1.

⁷ Average standard deviations of total group contributions for rounds 1-20 are 9.9, 5.3, 5.4, and 6.8 for NoMin, FixMin, RelMin, and ProgMin respectively.

⁸ Average standard deviations of total group contributions for round 1 are 9.0, 7.4, 6.9, and 7.9 for NoMin, FixMin, RelMin, and ProgMin respectively.



Figure 1: Average group contributions per round (by treatment)

Net relative group contributions and motivational crowding-out: To compare voluntary contributions in the NoMin treatment to the contributions in the MCS treatments, while taking into account the different sizes of strategy sets in these two kinds of treatments, we calculate for each round *t* the *net relative group contribution*:

$$Y_t^{Net} = \frac{\sum_{i=1}^4 y_{it} - \sum_{i=1}^4 c_i}{\sum_{i=1}^4 e_{it} - \sum_{i=1}^4 c_i} = \begin{cases} \frac{Y_t}{60} & \text{for NoMin} \\ \frac{Y_t - 24}{36} & \text{otherwise} \end{cases}$$
(2)

 Y_t^{Net} captures, for round *t*, the group contribution above the sum of mandatory contributions (group contribution minus sum of mandatory contributions) relative to the net endowment of the group (group endowment minus sum of mandatory contributions). This measure ranges from zero to one. It is zero, when only the mandatory levies are contributed (which are zero in the case of NoMin) and one, when the entire net endowment is contributed.

This measure allows a more flawless comparison of the treatments in our study since it takes into account that players with different endowments and different minimum contribution requirements have different strategy sets. Based on this measure we define a *"motivational crowding-out* or *crowding-in"* of voluntary contributions, which is different from the classic definition of crowding-out by public policy (Andreoni, 1993). If we observe that in a MCS treatment groups contribute a lower (higher) percentage of

their freely disposable endowments (net endowments) than the groups in the NoMin treatment, we interpret this observation as motivational crowding-out (crowding-in) of contributions by the minimum contribution requirements. Motivational crowding-out is, in contrast to the classic crowding-out, not measured token by token but in percent of the endowment that is at free disposal. While we define that net relative group contributions at the same level as in the NoMin treatment imply zero motivational crowding-out, full motivational crowding-out is defined by zero net relative group contributions. Both definitions are different from the classic definitions in Andreoni (1993).

Figure 2 shows the development of the average net relative group contributions in the four treatments. Visual inspection suggests two distinct contribution levels over rounds 1 to 20: NoMin and ProgMin show average contributions of 60.6 percent and 57.6 percent, respectively, while FixMin and RelMin show average contributions of 40.8 percent and 45.1 percent, respectively.^{9, 10} Pairwise comparisons, based on U tests, indicate significant differences between NoMin and FixMin as well as ProgMin and FixMin (p = .0963, for both comparisons). Thus, we find on average small and statistically insignificant motivational crowding-out of 4.9 percent in ProgMin and higher but still statistically insignificant motivational crowding-out of 25.5 percent in RelMin. Only the motivational crowding-out of 32.6 percent in FixMin is statistically significant.

Result 2: Motivational crowding-out of group contribution is statistically significant in FixMin. There is no significant motivational crowding-out in ProgMin and RelMin.

⁹ The corresponding average standard deviations of Y_t^{Net} for rounds 1-20 are .1645, .1478, .1503, and .1885, for NoMin, FixMin, RelMin, and ProgMin respectively.

¹⁰ The overall negative time trend is moderate and seems to be, except for the first periods, not different between the treatments.



Figure 2: Average net group contributions (by treatment)

4.2. Contributions by player types

4.2.1. Comparison within treatments

To compare the contributions of the different player types within each treatment, differences in the end-game effects for the player types play a minor role. We thus consider averages over rounds 1 to 25 in this analysis.

Figure A.1 in Appendix A presents the development of average contributions of the three player types (Type 10, Type 15, and Type 20) in the four treatments over the 25 rounds.¹¹ We find the non-surprising tendency for more abundantly endowed players to contribute more in absolute terms. Table A. 1 in Appendix A shows that, with the exception of two comparisons, all pairwise comparisons, based on signed-rank tests, show significant differences with $p \le .0593$.

Given the asymmetry in endowments, we consider two "relative contribution measures" to compare the contributions of the poor (Type 10), wealthy (Type15), and rich (Type 20) players. The first measure goes back to the fair-share rule in Hofmeyr et al. (2007) and Keser et al. (2014); it measures the absolute contributions of the player types

¹¹ For Type 10 (player 1) and Type 20 (player 4) the averages are based on ten players for each average, each. For Type 15 (players 2 and 3), the averages are based on twenty players.

relative to their individual endowment (relative contribution). Figure A. 2 in Appendix A presents the development of average relative contributions for the three player types in the four treatments. As can be seen, there are no significant differences in the average relative contributions between the player types both in NoMin (AsymWeak treatment of Keser et al. (2014)) and RelMin. For RelMin, where the mandatory contributions push contributions toward the "fair-share" rule with equal relative contributions, this result is highly plausible. In FixMin average relative contributions significantly differ between the player types such that relative contributions to the public good decrease with the endowment level. For ProgMin, on the other hand, the contribution hierarchy is reversed such that the wealthier players contribute relatively more. Table A. 2 in Appendix A provides the p-values of the pairwise comparisons of contributions by player type in the four treatments (signed-rank tests). These widely confirm the previous statement: all differences in NoMin and RelMin are insignificant, and (almost) all differences in FixMin and ProgMin are significant ($p \le .0745$, the unique exception is the difference between Types 15 and 20 in ProgMin that is not statistically significant). We may conclude that the proportional mandatory contributions in RelMin, the regressive mandatory contributions in FixMin, and the progressive mandatory contributions in ProgMin, respectively, exert their "intended" influence. RelMin leads to equal relative contributions, FixMin to higher relative contributions by the less wealthy players, and ProgMin to higher relative contributions by the more wealthy players.

Result 3: In RelMin players follow the simple "fair-share" rule of equal relative contributions of the endowment. However, this rule does not apply in FixMin and ProgMin: average relative contributions are higher for the less wealthy players in the regressive FixMin treatment and higher for the more wealthy players in the progressive ProgMin treatment. These results confirm Hypothesis 2.

The second relative-contribution measure, y_{it}^{Net} , essentially calculates Y_t^{Net} on an "individual" basis for each player type *i* ϵ {*Type* 10, *Type* 15, *Type* 20}:

$$y_{it}^{Net} = \frac{y_i - c_i}{e_i - c_i},$$
(3)

where c_i is treatment dependent. It captures for each player type *i* the average individual contribution net of the mandatory contribution (absolute contribution minus mandatory contribution) relative to the net endowment (individual endowment minus mandatory contribution).

Figure A. 3 in Appendix A presents the development of average y_{it}^{Net} for the three player types in the four treatments. Table A. 3 in Appendix A shows the p-values of the pairwise comparisons based on signed-rank tests. As can be seen, there is only one difference that is statistically significant (p = .0745, Type 10 players contribute significantly more than Type 20 players in FixMin). None of the remaining comparisons yields statistical significance. We conclude that players tend to follow a modified "net fair-share" rule of equal relative contributions of the disposable endowment in FixMin and ProgMin. The

"net fair-share" rule also applies in NoMin and RelMin, since it coincides with the original "fair-share" rule.

Result 4: *Players follow in all treatments a "net fair-share" rule of equal contributions relative to the decision space.*

4.2.2. Comparison between treatments

So far, we have focused our analysis on the differences between the player types within each of the four treatments. For a deeper understanding of the mechanics that might be at work in the various treatments, i.e., how the norms might be set by the different MCS regimes, we compare the behavior of each player type across the four treatments. If we observe differences in behavior, these differences could ultimately lead to differences in group contributions between the treatments. Due to the strong end-game effect in NoMin, which is impeded by the minimum contribution requirements in the MCS treatments, we report only averages over rounds 1 to 20 in this analysis.

Figure 3 depicts that, on average, relative contributions (and thus also absolute contributions) of Type 10 are highest in FixMin, where also the mandatory minimum contribution relative to the endowment is highest for this player type. The trajectories in the other treatments are not clearly distinguishable. Contributions of Type 15 are clearly highest in ProgMin and lowest in NoMin; the trajectories in FixMin and RelMin lie in between. Contributions of Type 20 are highest in ProgMin and lowest in FixMin. NoMin shows a clear decline that, toward the end of the game, even undercuts the level of FixMin. Contributions in RelMin lie between those in FixMin and ProgMin.

Considering contributions in the first round, we find that, on average, Type 10 players contribute 6.3, 7.2, 6.4, and 5.7; Type 15 players 6.9, 9.4, 9.3, and 9.7; and Type 20 players 10.7, 13, 11.6, and 13.5 in NoMin, FixMin, RelMin, and ProgMin, respectively. For Types 10 and 20, the differences between the treatments are never statistically significant. For Type 15, differences are significant between NoMin and all three MCS treatments ($p \le .0698$, U tests); comparisons between the MCS treatments yield no significant differences.



Figure 3: Average relative contributions (by treatment)

For contributions in rounds 1 to 20, Table A. 4 in Appendix A shows the p-values of the pairwise comparisons of average absolute [and net] contributions for the three player types (U tests). As can be seen, Type 10 players contribute on average most in FixMin and least in ProgMin and NoMin; contributions in RelMin lie in between. However, only the difference between ProgMin and FixMin is statistically significant (p = .0257). Type 15 players contribute in all MCS treatments on average more than in NoMin but only the difference between ProgMin and NoMin is significant (p = .0232). Although differences between the MCS treatments are not significant, the average figures indicate that contributions are highest in ProgMin, followed by RelMin, and then FixMin. Type 20 players contribute on average least in FixMin and most in ProgMin; contributions in NoMin and RelMin lie in between. Contributions in ProgMin are thereby significantly higher than in all other treatments ($p \le .0696$), between which there is no statistically significant difference. For Type 20 we observe motivational crowding-out in FixMin and RelMin ($p \le .0696$, for both). For no other player type do we observe motivational crowding-out. To conclude, for player types 15 and 20 but not for player type 10, ProgMin leads to the highest contributions to the public good. Given that Type 10 players have a lower leverage on group contributions than Type 15 and Type 20 players,

this explains why we observe the highest group contribution level in ProgMin, which is significantly higher than in NoMin.

Result 5: Type 10 and Type 20 players contribute most, when they are facing relatively high mandatory contributions (FixMin and ProgMin, respectively) and contribute least, when they are facing relatively low mandatory contributions (ProgMin and FixMin, respectively). Type 15 players contribute most in ProgMin and least in NoMin.

One might argue that contributions in the MCS treatments increase merely due to the enforced increase in the contributions of uncooperative subjects, while the cooperative subjects' contributions might have remained the same. To test for this eventuality, we use a simple approach. For each player type and treatment, we order average contributions from the lowest to the highest, and divide this ordering by half. We can thus distinguish between more and less cooperative subjects and compare the behavior of the more cooperative subjects' in the various treatments. The average contributions are presented in Table A. 5 in Appendix A. Tests for differences across the treatments, based on U tests, confirm for both Type 15 and 20 players that the increase in contributions in ProgMin is not solely driven by the higher contributions of uncooperative subjects.¹²

Additionally, we find that in ProgMin lower-bound contributions of Type 15 players $(y_i = c_i = 6)$, for which mandatory contributions are equal in all MCS treatments, exhibit with 19.8 percent the lowest proportion of all MCS treatments ($p \le .0030$, U tests). In NoMin, we observe 38.3 percent of contributions of six or below. Furthermore, we also find that in ProgMin Type 15 and 20 players display the highest proportion of full contributions ($y_i = e_i = 15$ and $y_i = e_i = 20$, respectively) compared to all other treatments ($p \le .0225$, U tests); for Type 10 players differences in the proportions of full contributions between the treatments are insignificant. The respective figures for lower-bound and full contributions are presented in Table A. 6 in Appendix A.

The results concerning contributions by cooperative players and concerning lower bound and full contributions indicate that the ProgMin treatment leads to a norm shift, not just to higher group contributions due to a higher constraint for the wealthier players.

¹² Cooperative Type 15 players significantly contribute more in ProgMin than in RelMin and FixMin ($p \le .0413$); all other differences are not significant. Cooperative Type 20 players contribute significantly more in ProgMin than in NoMin and FixMin (p = .0413, for both comparisons); all other differences are not significant. For Type 10 players, contributions of cooperative players are not significantly different between all treatments.

4.3. Reciprocity

As known from the literature (e.g., Keser and van Winden, 2000; Fischbacher et al., 2001), subjects behave reciprocally and make contributions that depend on the actions of others. Keser and van Winden (2000) argue that players, if they change their contribution from one round to the next, tend to increase (decrease) their contribution, if in the previous round their contribution was below (above) the average of the group. Given the different endowments of player types in our experiment, we examine reciprocity in terms of changes in relative contributions. In other words, we examine how players react if their own relative contribution has been lower (higher) than the average of the relative contributions of the other group members in the previous round.

Table 2 presents the results of a regression on the reaction of the individual relative contribution to the average relative contribution of the other group members in the previous round. The dependent variable is the individual change of the relative contribution to the public investment from the previous to the current round (Δy_{it}^{Rel}) . The major explanatory variable is the lagged difference between the relative contribution of the player and the average relative contribution of the other group members (*L. Diff 2MeanOthers*). Additionally, we control for player-type and treatment effects by using dummy variables (and interactions), considering Type 10 and NoMin as the respective reference group.

Table 2 indicates that *L*. *Diff2MeanOthers* is significantly negative and thus provides clear evidence for reciprocity. Another important finding concerns the ProgMin treatment and potentially explains why it shows the highest group contribution level. On the one hand, we observe that Type 10 players behave most reciprocally (the interaction term *ProgMin × L.Diff2MeanOthers* is significantly negative) and, on the other hand, that Type 15 and Type 20 players behave least reciprocally (the interaction terms *Type 15* × *ProgMin* × *L.Diff2MeanOthers* and *Type 20* × *ProgMin* × *L.Diff2MeanOthers* are significantly positive). These findings are consistent with our previous results. As it seems, Type 15 and Type 20 players choose their contributions in this treatment "irrespective" of the contributions by Type 10 players. The two wealthier types seem to accept the low contributions by Type 10 players and "simply follow" the contribution norms set by the ProgMin schedule. This may, at least partially, explain why contributions of both player types in ProgMin are highest in this treatment, despite the fact that Type 10 players exhibit their lowest contributions of all treatments. Additionally, we find that the change in relative contributions is becoming more negative during the course of the game (*Round* is significantly negative).

Result 6: *Players generally behave reciprocally. In ProgMin, Type 10 players behave more and Type 15 and 20 players less reciprocally.*

Table 2:

R	egression	for	comparisons	of	recip	rocity
τ,	CSI COSIOII	101	comparisons	O1	recip	rocity

Dependent variable:	Δy_i^I	Rel t
Constant	.0378	(.0251)
L.Diff2MeanOthers	3268***	(.0552)
Round	0021***	(.0007)
Type 15	0390	(.0291)
Type 20	0043	(.0295)
FixMin	.0437	(.0288)
RelMin	.0201	(.0294)
ProgMin	0855**	(.0340)
Type 15 × FixMin	0267	(.0343)
Type 15 × RelMin	0001	(.0352)
Type 15 × ProgMin	.1312***	(.0389)
Type 20 × FixMin	1372***	(.0387)
Type 20 × RelMin	0417	(.0367)
Type 20 × ProgMin	.0981**	(.0398)
<i>Type 15 × L.Diff2MeanOthers</i>	1285*	(.0703)
Type 20 × L.Diff2MeanOthers	0313	(.0823)
FixMin × L.Diff2MeanOthers	.0601	(.0765)
RelMin × L.Diff2MeanOthers	0355	(.0800)
ProgMin × L.Diff2MeanOthers	2083**	(.0849)
Type 15 × FixMin × L.Diff2MeanOthers	.0072	(.0991)
Type 15 × RelMin × L.Diff2MeanOthers	0576	(.1030)
Type 15 × ProgMin × L.Diff2MeanOthers	.1995*	(.1078)
Type 20 × FixMin × L.Diff2MeanOthers	1446	(.1244)
Type 20 × RelMin × L.Diff2MeanOthers	0840	(.1183)
Type 20 × ProgMin × L.Diff2MeanOthers	.3226***	(.1151)
R^2	.23	46

Notes: OLS-regressions with robust variance estimates. Standard errors in parentheses. *, **, and *** indicate significance at the 10%, 5%, and 1% level, respectively.

3040

4.4. Profits and Gini indices

Ν

Table 3 illustrates the average profits per round over rounds 1 to 20 for the three player types and the resulting average group profits in the four treatments. Pairwise comparisons, using U tests, show that solely the differences in average group profits between ProgMin and NoMin (p = .0343), and ProgMin and FixMin (p = .0963) are significant.

	NoMin	FixMin	RelMin	ProgMin
Type 10	43.30	42.30	45.60	52.23
Type 15	49.33	49.03	50.31	52.04
Type 20	50.75	57.07	54.27	53.19
Group profit	192.70	197.41	200.48	209.49

Table 3:Average profits per round

Notes: Averages for rounds 1 to 20. Social optimum sum of profits per round = 240. Equilibrium sum of profits per round = 120, 168, 168, 168. Equilibrium profit per round for Type 1 = 20, 32, 36, 40; equilibrium profit per round for Type 2 = 30, 42, 42, 42; equilibrium profit per round for Type 3 = 40, 52, 48, 44 (NoMin, FixMin, RelMin, ProgMin). Social optimum profit per round = 60 (for all types).

Comparisons for the player types between treatments based on U tests show that Type 10 players earn on average significantly more in ProgMin than in all other treatments, between which there are no significant differences ($p \le .0343$). For Type 15 players, profits in ProgMin are significantly higher than in NoMin (p = .0989) and FixMin (p = .0498); for the other differences, we cannot reject the null hypothesis of no significant difference. For Type 20 players, profits in FixMin are significantly higher than in NoMin and ProgMin ($p \le .0172$); all other differences are statistically insignificant.

Given that players in all four treatments start with different endowments, and that contributions to the public investment tend to result in an equalization of total profits through a redistribution of wealth, we analyze, based on the Gini index, the differences in inequality between the initial distribution of endowments, the distribution of total profits in equilibrium, and the actual distribution of total profits over rounds 1 to 20 in the four treatments. Table 4 displays the respective figures.

Average total profit Gini indices are smaller than Nash-equilibrium Gini indices for NoMin (p = .0051) and FixMin (p = .0284), using MPSR tests; for RelMin the difference is almost significant (p = .1141). For ProgMin, on the other hand, total profit Gini indices are significantly higher than Gini indices in equilibrium (p = .0218).

Furthermore, the comparison of average total profit Gini indices discloses that ProgMin exhibits significantly smaller Gini indices than all other treatments. Based on U tests, we find significant differences between NoMin and ProgMin, and FixMin and RelMin (p = .0284, respectively), and also between FixMin and ProgMin, and RelMin and ProgMin (p = .0065, respectively).

Tab	le 4:
Gini	indices

Treatment	Gini index for endowments [net of MCs] (1)	Gini index in Nash equillibrium (2)	Gini index for total profits (3)	Differences in percent (3)-(1)/ (2)-(1)/(3)-(2)
NoMin	.1250 [.1250]	.1250	.0561	-55.1/-0.0/-55.1
FixMin	.1250 [.2083]	.0893	.0674	-46.1/-28.6/-24.5
RelMin	.1250 [.1250]	.0536	.0471	-62.3/-57.1/-12.1
ProgMin	.1250 [.0417]	.0179	.0279	-77.7/-85.7/+55.9

Notes: Gini indices for total profits over rounds 1 to 20.

Result 7: *ProgMin leads to the lowest inequality in total profits of all treatments. Although, the inequality in total profits is larger than in equilibrium, the average group profit in ProgMin is the highest of all treatments.*

5. Conclusion

We investigate whether and how cooperation and the previously observed "fair-share" norm in public-good experiments with asymmetrically endowed players are influenced by enforced minimum-contribution schedules. We consider schedules, where all players face the same absolute minimum contribution irrespective of their endowment (FixMin), where all players face the same minimum contribution relative to the endowment (RelMin), and where a player with a higher endowment faces a higher minimum contribution relative to the endowment than a player with a lower endowment (ProgMin). Our mandatory minimum-contribution schedules relate to the literature on tax fairness or "vertical equity". In taxpayer surveys, Gerbing (1988) and Roberts and Hite (1994) find evidence of a preference for progressive tax rates. For upper-income taxpayers, however, Gerbing finds that they perceive flat tax rates as more fair. In the context of a public-good game, where participants can vote for several minimum contribution schemes, which are intended to provide a jointly agreed minimum group provision level, Gallier et al. (2014) find that the scheme which equalizes payoffs (similar to ProgMin) is mostly chosen by less wealthy players, while rich players mostly chose the scheme which equalizes contributions (similar to FixMin). Given this evidence and the pervasive calls for fairer tax systems implying tax breaks for lower and middle income classes together with tax increases for upper income classes, it is possible that an as fairer perceived distribution of mandatory minimum contributions (as, for

example, in ProgMin) exerts a positive effect on individual and consequently total group contributions.

The results of our experiment suggest the potential of mandatory minimum contributions to exert expressive power. We observe them to exert a norm-giving character. They seem to communicate relations of fair contributions by the different player types and thus might increase group contributions relative to the situation without minimum-contribution requirements. It turns out that this is particularly true for our ProgMin treatment, which is likely perceived as the most fair among all mandatory contribution systems considered. ProgMin is the only treatment, where the crowding-out of voluntary contributions to the public good by mandatory contributions is significantly incomplete, when we use the measure by Andreoni (1993). For RelMin and FixMin the crowding-out is close to complete. When we consider motivational crowding-out as defined in this paper, it is statistically significant only in FixMin. ProgMin exhibits hardly any motivational crowding-out.

On the individual level, we find support for the "fair-share" rule in RelMin. This rule cannot be detected in FixMin and ProgMin due to the norms set through the (inverted) progressivity in both treatments. In the regressive FixMin treatment average relative contributions are higher for less wealthy players and in ProgMin average relative contributions are higher for more wealthy players. As we see, the "fair-share" norm can be eroded through a deliberate intervention. In particular, in ProgMin, the norm of what is a player's fair share is adapted in the "intended" direction. Players in FixMin and ProgMin seem to coordinate on a modified fair-share rule of equal contributions relative to the decision space, which we call the "net fair-share" rule. Average relative contributions to the available decision space are equal for all player types in FixMin and in ProgMin.

Type 15 players, for which mandatory contributions are the same in all three MCS treatments, contribute most in ProgMin and least in NoMin. The other two player types contribute more, when they are facing relatively high mandatory contributions (FixMin for Type 10 and ProgMin for Type 20) and contribute less, when they are facing relatively low mandatory contributions (ProgMin for Type 10 and FixMin for Type 20). We also find Type 10 players to behave most and Type 15 and Type 20 players to behave least reciprocally in ProgMin. As it seems, Type 15 and Type 20 players choose their contributions in this treatment "irrespective" of the lower contributions by Type 10 players. This may, at least partially, explain why contributions of both player types in ProgMin are the highest of all treatments, despite the fact that Type 10 players exhibit their lowest contributions of all treatments. Thus, the observation that group contributions are significantly higher in ProgMin than in the other three treatments can potentially be explained by the acceptance of the norm of progressive contributions among the Type 15 and Type 20 players in this treatment, rendering their contributions unconditional on the contributions by Type 10 players. Furthermore, we find that ProgMin exhibits the lowest inequality in total profits of all our treatments in terms of the Gini index.

In spite of these strong results, we advise caution generalizing our findings, in particular with respect to public policy. The response of contributions in a public-good game with heterogeneous endowments to mandatory minimum contributions may not be the same as the response of real economic factors as, for example, labor supply on an intervention in this sphere (Lindsey, 1987). In our experiment, heterogeneous endowments were randomly allocated to all participants in a group. Thus, participants neither had to supply their endowments by themselves nor to work for them. Even though neither Clark (2002) nor Cherry et al. (2005) find that these origins of the endowments, compared to cases where participants are provided with windfall endowments, have an effect in their public-good experiments, we believe that at least some caution is advised concerning possible effects of the endowment origin. This might be particularly true, when the asymmetry in the endowment distribution becomes more important. For example, Cherry et al. (2002) find an effect of endowment origin on behavior in a dictator-game setting. If endowments had to be earned, our mandatory contributions could likely exert similar effects as taxes and lead to a decline in the work effort, which would be in keeping with the Laffer curve hypothesis. Note, however, that for almost all types (with the exception of Type 10 in FixMin and Type 20 in ProgMin) our tax rates are well below the empirically observed tax-revenue-maximizing rates of 50 to 60 percent (e.g., Sutter and Weck-Hannemann, 2003). Although our study is able to show that the progressive minimum-contribution schedule performed best in our public-good setting in terms of overall contribution rates, we are not able to predict, which degree of progression would work best in a public-good environment, where endowments must be earned.

With respect to our initial example of public institutions, which rely on two-tier financing models based on mandatory admission charges plus voluntarily charitable donations and/or employ third degree price discrimination by setting admission fees that vary by status (e.g., regular tickets and reduced tickets for children, students, retirees, unemployed etc.), the increase in average group contributions in ProgMin compared to NoMin and FixMin suggests that progressive tariff structures can indeed be used to improve the financing of such institutions.

Acknowledgements

We thank Emmanuel Peterle and Holger Rau for helpful comments. Financial support by Deutsche Forschungsgemeinschaft (DFG) GRK 1666 GlobalFood is gratefully acknowledged.

References

- Abrams, B.A., Schmitz, M.D., 1978. The 'Crowding-out' Effect of Governmental Transfers on Private Charitable Contributions. Public Choice 33 (1), 29-39.
- Abrams, B.A., Schmitz, M.D., 1984. The Crowding-out Effect of Governmental Transfers on Private Charitable Contributions: Cross-Section Evidence. National Tax Journal 37 (4), 563–568.
- Anderson, S.P., Goeree, J.K., Holt, C.A., 1998. A Theoretical Analysis of Altruism and Decision Error in Public Goods Games. Journal of Public Economics 70 (2), 297–323.
- Andreoni, J., 1993. An Experimental Test of the Public-Goods Crowding-Out Hypothesis. The American Economic Review 83 (5), 1317–1327.
- Andreoni, J., Payne, A.A., 2011. Is Crowding Out Due Entirely to Fundraising? Evidence from a Panel of Charities. Journal of Public Economics (Charitable Giving and Fundraising Special Issue) 95 (5–6), 334–343.
- Bergstrom, T., Blume, L., Varian, H.R., 1986. On the private provision of public goods. Journal of Public Economics 29 (1), 25–49.
- Bernheim, B.D., 1986. On the Voluntary and Involuntary Provision of Public Goods. The American Economic Review 76 (4), 789–793.
- Brekke, K.A., Kverndokk, S., Nyborg, K., 2003. An Economic Model of Moral Motivation. Journal of Public Economics 87 (9–10), 1967–1983.
- Chan, K.S., Godby, R., Mestelman, S., Muller, R.A., 2002. Crowding-out Voluntary Contributions to Public Goods. Journal of Economic Behavior & Organization 48 (3), 305–317.
- Cherry, T.L., Frykblom, P., Jason F. Shogren, 2002. Hardnose the Dictator. The American Economic Review 92 (4), 1218–1221.
- Cherry, T.L., Kroll, S., Shogren, J.F., 2005. The Impact of Endowment Heterogeneity and Origin on Public Good Contributions: Evidence from the Lab. Journal of Economic Behavior & Organization 57 (3), 357–365.
- Clark, J., 2002. House Money Effects in Public Good Experiments. Experimental Economics 5 (3), 223-231.
- Condry, J., Chambers, J., 1978. Intrinsic Motivation and the Process of Learning, in: Lepper, M.R., Greene, D.((Eds.), The Hidden Costs of Reward: New Perspectives on the Psychology of Human Motivation. John Wiley, New York, pp. 61–84.
- Cooter, R., 1998. Expressive Law And Economics. The Journal of Legal Studies 27 (S2), 585–607.
- Davis, D.D., Holt, C.A., 1993. Experimental Economics. Princeton University Press, Princeton, New Jersey.
- Dawes, R.M., Thaler, R.H., 1988. Anomalies: Cooperation. The Journal of Economic Perspectives 2 (3), 187–197.
- Deci, E.L., 1971. Effects of Externally Mediated Rewards on Intrinsic Motivation. Journal of Personality and Social Psychology 18 (1), 105–115.
- Deci, E.L., 1972. Intrinsic Motivation, Extrinsic Reinforcement, and Inequity. Journal of Personality and Social Psychology 22 (1), 113–120. 10.1037/h0032355.

- Deci, E.L., Koestner, R., Ryan, R.M., 1999. A Meta-Analytic Review of Experiments Examining the Effects of Extrinsic Rewards on Intrinsic Motivation. Psychological Bulletin 125 (6), 627–668. 10.1037/0033-2909.125.6.627.
- Falk, A., Kosfeld, M., 2006. The Hidden Costs of Control. The American Economic Review 96 (5), 1611–1630.
- Fischbacher, U., Gächter, S., Fehr, E., 2001. Are People Conditionally Cooperative? Evidence from a Public Goods Experiment. Economics Letters 71 (3), 397–404.
- Fischbacher, U., 2007. z-Tree: Zurich Toolbox for Ready-made Economic Experiments. Experimental Economics 10 (2), 171–178.
- Frey, B.S., 1993. Does Monitoring Increase Work Effort? The Rivalry with Trust and Loyalty. Economic Inquiry 31 (4), 663–670.
- Frey, B.S., 1994. How Intrinsic Motivation is Crowded Out and In. Rationality and Society 6 (3), 334–352.
- Frey, B.S., Oberholzer-Gee, F., Eichenberger, R., 1996. The Old Lady Visits Your Backyard: A Tale of Morals and Markets. Journal of Political Economy 104 (6), 1297–1313.
- Frey, B.S., 1997a. A Constitution for Knaves Crowds out Civic Virtues. The Economic Journal 107 (443), 1043–1053.
- Frey, B.S., 1997b. Not Just for the Money: An Economic Theory of Personal Motivation. Edward Elgar Publishing, Cheltenham, UK and Brookfield, USA.
- Frey, B.S., Goette, L., 1999. Does Pay Motivate Volunteers? Working Paper Series No. 7. Institute for Empirical Research in Economics, University of Zurich.
- Frey, B.S., Jegen, R., 2001. Motivation Crowding Theory. Journal of Economic Surveys 15 (5), 589–611.
- Frey, B.S., Oberholzer-Gee, F., 1997. The Cost of Price Incentives: An Empirical Analysis of Motivation Crowding- Out. The American Economic Review 87 (4), 746–755.
- Galbiati, R., Vertova, P., 2008. Obligations and Cooperative Behaviour in Public Good Games. Games and Economic Behavior 64 (1), 146–170.
- Galbiati, R., Vertova, P., 2014. How Laws Affect Behavior: Obligations, Incentives and Cooperative Behavior. International Review of Law and Economics 38, 48–57.
- Gallier, C., Kesternich, M., Sturm, B., 2014. Voting for Burden Sharing Rules in Public Goods Games. ZEW Discussion Paper No. 14-056.
- Gerbing, M.D., 1988. An Empirical Study of Taxpayer Perceptions of Fairness. Unpublished PhD Thesis, University of Texas, Austin, USA.
- Gneezy, U., Rustichini, A., 2000a. A Fine Is a Price. The Journal of Legal Studies 29, 1–17.
- Gneezy, U., Rustichini, A., 2000b. Pay Enough or Don't Pay at All. Quarterly Journal of Economics 115 (3), 791–810. 10.1162/003355300554917.
- Greiner, B., 2004. The Online Recruitment System ORSEE 2.0 A Guide for the Organization of Experiments in Economics. University of Cologne, Working Paper Series in Economics 10 (23), 63–104.
- Hofmeyr, A., Burns, J., Visser, M., 2007. Income Inequality, Reciprocity and Public Good Provision: An Experimental Analysis. South African Journal of Economics 75 (3), 508–520.

- Keser, C., 1996. Voluntary Contributions to a Public Good when Partial Contribution is a Dominant Strategy. Economics Letters 50 (3), 359–366.
- Keser, C., Markstädter, A., Schmidt, M., Schnitzler, C., 2014. Social Costs of Inequality -Heterogeneous Endowments in Public-Good Experiments. cege Discussion Paper 217, University of Göttingen.
- Keser, C., Schmidt, M., 2014. Cooperation and Punishment among Heterogeneous Agents. Working Paper, University of Göttingen.
- Keser, C., van Winden, F., 2000. Conditional Cooperation and Voluntary Contributions to Public Goods. The Scandinavian Journal of Economics 102 (1), 23–39. 10.1111/1467-9442.00182.
- Ledyard, J.O., 1995. Public Goods: A Survey of Experimental Research, in: Kagel, J.H., Roth, A.E. (Eds.), The Handbook of Experimental Economics. Princeton University Press, New Jersey.
- Lepper, M.R., Greene, D., Nisbett, R.E., 1973. Undermining Children's Intrinsic Interest with Extrinsic Reward: A Test of the "Overjustification" Hypothesis. Journal of Personality and Social Psychology 28 (1), 129–137.
- Lindsey, L.B., 1987. Individual Taxpayer Response to Tax Cuts: 1982–1984: With Implications for the Revenue Maximizing Tax Rate. Journal of Public Economics 33 (2), 173–206.
- Marwell, G., Ames, R.E., 1979. Experiments on the Provision of Public Goods. I. Resources, Interest, Group Size, and the Free-rider Problem. American Journal of Sociology, 1335–1360.
- Marwell, G., Ames, R.E., 1980. Experiments on the Provision of Public Goods. II. Provision Points, Stakes, Experience, and the Free-rider Problem. American Journal of Sociology, 926–937.
- McAdams, R.H., Nadler, J., 2005. Testing the Focal Point Theory of Legal Compliance: The Effect of Third-Party Expression in an Experimental Hawk/Dove Game. Journal of Empirical Legal Studies 2 (1), 87–123.
- Nyborg, K., Rege, M., 2003. Does Public Policy Crowd Out Private Contributions to Public Goods. Public Choice 115 (3-4), 397-418.
- Ostrom, E., 2000. Collective Action and the Evolution of Social Norms. The Journal of Economic Perspectives 14 (3), 137–158.
- Payne, A.A., 1998. Does the Government Crowd-out Private Donations? New Evidence from a Sample of Non-Profit Firms. Journal of Public Economics 69 (3), 323–345.
- Riedel, N., Schildberg-Hörisch, H., 2013. Asymmetric Obligations. Journal of Economic Psychology 35 (0), 67–80.
- Roberts, M.L., Hite, P.A., 1994. Progressive Taxation, Fairness, and Compliance. Law & Policy 16 (1), 27–48.
- Roberts, R.D., 1984. A Positive Model of Private Charity and Public Transfers. The Journal of Political Economy, 136–148.
- Roberts, R.D., 1987. Financing Public Goods. Journal of Political Economy 95 (2), 420–437.

- Rotter, J.B., 1966. Generalized Expectancies for Internal versus External Control of Reinforcement. Psychological Monographs: General and Applied 80 (1), 1–28 (whole No. 609).
- Ryan, R.M., Deci, E.L., 2000. Intrinsic and Extrinsic Motivations: Classic Definitions and New Directions. Contemporary Educational Psychology 25 (1), 54–67.
- Sutter, M., Weck-Hannemann, H., 2003. Taxation and the Veil of Ignorance A Real Effort Experiment on the Laffer Curve. Public Choice 115 (1-2), 217-240. 10.1023/A:1022873709156.
- Titmuss, R.M., 1970. The Gift Relationship: From Human Blood to Social Policy. Allen & Unwin (London), London.
- Uptom, W.E., 1973. Altruism, Attribution, and Intrinsic Motivation in the Recruitment of Blood Donors, Cornell University.
- Warr, P.G., 1982. Pareto Optimal Redistribution and Private Charity. Journal of Public Economics 19 (1), 131–138.
- Warr, P.G., 1983. The Private Provision of a Public Good is Independent of the Distribution of Income. Economics Letters 13 (2–3), 207–211.

Appendix A

Additional Tables and Figures



Figure A. 1: Average absolute contributions by player type

Table A. 1:

Comparisons of average absolute contributions between player types (p-values of two-sided signed-rank tests)

		NoMin				FixMin	
	Type 10 (6.31)	Type 15 (7.65)	Type 20 (11.44)		Type 10 (8.03)	Type 15 (9.54)	Type 20 (10.56)
Type 10 (6.31)	-	.2845	.0218	Type 10 (8.03)	-	.0593	.0284
Type 15 (7.65)	-	-	.0051	Type 15 (9.54)	-	-	.2411
Type 20 (11.44)	-	-	-	Type 20 (10.56)	-	-	-
		D - Mi-					
		Reimin				ProgMin	
	Туре 10 (7.18)	Type 15 (9.82)	Type 20 (12.92)		Type 10 (6.07)	ProgMin Type 15 (11.00)	Type 20 (15.48)
Type 10 (7.18)	Type 10 (7.18) -	Type 15 (9.82) .0166	Type 20 (12.92) .0069	Type 10 (6.07)	Type 10 (6.07) -	ProgMin Type 15 (11.00) .0051	Type 20 (15.48) .0051
Type 10 (7.18) Type 15 (9.82)	Type 10 (7.18) - -	Type 15 (9.82) .0166	Type 20 (12.92) .0069 .0069	Type 10 (6.07) Type 15 (11.00)	Type 10 (6.07) -	ProgMin Type 15 (11.00) .0051 -	Type 20 (15.48) .0051 .0051

Notes: Comparisons and average contributions involve rounds 1 to 25. Averages for the player type are given in parentheses.



Figure A. 2: Average relative contributions by player type

Table A. 2:

Comparisons of average relative contributions between player types (p-values of two-sided signed-rank tests)

		NoMin				FixMin	
	Type 10 (.6308)	Type 15 (.5101)	Type 20 (.5720)		Type 10 (.8032)	Type 15 (.6359)	Type 20 (.5280)
Type 10 (.6308)	-	.2411	.6465	Type 10 (.8032)	-	.0166	.0093
Type 15 (.5101)	-	-	.1394	Type 15 (.6359)	-	-	.0593
Type 20 (.5720)	-	-	-	Type 20 (.5280)	-	-	-
		RelMin				ProgMin	
	Type 10 (.7184)	RelMin Type 15 (.6549)	Type 20 (.6460)		Type 10 (.6068)	ProgMin Type 15 (.7332)	Type 20 (.7740)
Type 10 (.7184)	Type 10 (.7184) -	RelMin Type 15 (.6549) .1394	Type 20 (.6460) .2845	Type 10 (.6068)	Type 10 (.6068) -	ProgMin Type 15 (.7332) .0593	Type 20 (.7740) .0745
Type 10 (.7184) Type 15 (.6549)	Type 10 (.7184) - -	RelMin Type 15 (.6549) .1394	Type 20 (.6460) .2845 .2411	Type 10 (.6068) Type 15 (.7332)	Type 10 (.6068) - -	ProgMin Type 15 (.7332) .0593	Type 20 (.7740) .0745 .3329

Notes: Comparisons and average contributions involve rounds 1 to 25. Averages for the player type are given in parentheses.



Figure A. 3: Average net contributions by player type

Table A. 3:

Comparisons of average net contributions between player types (p-values of two-sided signed-rank tests)

		NoMin				FixMin	
	Type 10 (.6308)	Type 15 (.5101)	Type 20 (.5720)		Type 10 (.5080)	Type 15 (.3931)	Type 20 (.3257)
Type 10 (.6308)	-	.2411	.6465	Type 10 (.5080)	-	.1394	.0745
Type 15 (.5101)	-	-	.1394	Type 15 (.3931)	-	-	.5076
Type 20 (.5720)	-	-	-	Type 20 (.3257)	-	-	-
		RelMin				ProgMin	
	Type 10 (.5307)	RelMin Type 15 (.4249)	Type 20 (.4100)		Type 10 (.5085)	ProgMin Type 15 (.5553)	Type 20 (.5480)
Type 10 (.5307)	Type 10 (.5307) -	RelMin Type 15 (.4249) .1394	Type 20 (.4100) .2845	Type 10 (.5085)	Type 10 (.5085) -	ProgMin Type 15 (.5553) .5076	Type 20 (.5480) .7213
Type 10 (.5307) Type 15 (.4249)	Type 10 (.5307) - -	RelMin Type 15 (.4249) .1394	Type 20 (.4100) .2845 .2411	Type 10 (.5085) Type 15 (.5553)	Type 10 (.5085) - -	ProgMin Type 15 (.5553) .5076	Type 20 (.5480) .7213 .9594

Notes: Comparisons and average contributions involve rounds 1 to 25. Averages for the player type are given in parentheses.

Table A. 4:

By Type: Comparisons of average absolute [and average net] contributions between treatments (p-values of two-sided U tests)

		Type 10		
	NoMin	FixMin	RelMin	ProgMin
NoMin		.1211	.5453	.7624
INOIVIIII	-	[.5453]	[.4963]	[.4057]
FixMin			.3445	.0257
FIXIVIIII	-	-	[1.000]	[.8205]
PolMin		_	_	.2725
Kellwiili	-	-	-	[.9097]
ProgMin	-	-	-	-
		Type 15		
	NoMin	FixMin	RelMin	ProgMin
NoMin		.1617	.1508	.0232
INOIMIII	-	[.1988	[.2265]	[.8798]
FivMin	_	_	.8798	.1304
I IXIVIIII		_	[.8798]	[.1306]
RelMin	_	-	-	.1123
Rentim				[.1124]
ProgMin	-	-	-	-
		Type 20		
	NoMin	FixMin	RelMin	ProgMin
NoMin		.2265	.9698	.0639
INOIMIII	-	[.0191]	[.0696]	[.5706]
FivMin	_	_	.1509	.0082
I IXIVIIII		_	[.4057]	[.0963]
RelMin	_	_	-	.0696
I CHMIII				[.1988]
ProgMin	-	-	-	-

Note: Average contributions over rounds 1 to 20 in NoMin, FixMin, RelMin, ProgMin: Type10: \bar{y}_i : 6.53, 8.21, 7.32, 6.26; \bar{y}_i^{Net} : .6525, .5513, .5533, .5325. Type 15: \bar{y}_i : 8.51, 9.84, 9.97, 11.35; \bar{y}_i^{Net} : .5675, .4267, .4408, .5947. Type 20: \bar{y}_i : 12.80, 10.82, 12.99, 15.78; \bar{y}_i^{Net} : .6400, .3443, .4154, .5780.

Table A. 5:

		NoMin	FixMin	RelMin	ProgMin
	U	4.4	7.4	6.0	4.6
Type 10	С	8.6	9.0	8.7	7.9
	U	5.6	7.7	8.1	9.5
Type 15	С	11.4	12.0	11.8	13.2
	U	10.0	7.7	10.3	13.5
Type 20	С	15.6	13.9	15.6	18.1

Comparisons of contributions by cooperative and uncooperative players within treatments

Notes: Average contribution figures involve rounds 1 to 20. U = Uncooperative players, C = Cooperative players. For Type 10 and 20 there are per definition respectively 5 U and 5 C players per treatment and for Type 15 respectively 10 U and 10 C players per treatment.

Table A. 6:

Relative frequency of individual decisions at the lower bound or full contribution to the public investment

	Lower-bound contributions (in percent)	Full contributions (in percent)
NoMin – Type 10	15.0 [39.5; 25.0; 17.5]	41.5
NoMin – Type 15	14.3 [38.3; 38.3; 38.3]	32.3
NoMin – Type 20	10.0 [19.0; 23.0; 27.0]	32.0
FixMin – Type 10	32.0	44.0
FixMin – Type 15	33.3	24.8
FixMin – Type 20	28.0	17.5
RelMin – Type 10	20.5	37.0
RelMin – Type 15	28.8	18.8
RelMin – Type 20	18.0	17.5
ProgMin – Type 10	25.5	36.5
ProgMin – Type 15	19.8	40.0
ProgMin – Type 20	25.0	43.5

Notes: All figures involve rounds 1 to 20. Lower bound contributions are 0 in NoMin for all player types; 6 in FixMin for all player types; 4, 6, and 8 in RelMin; and 2, 6, and 10 in ProgMin for Type 10, Type 15, and Type 20 players, respectively. Figures in [] respectively show the percentage of contributions that were below the minimum contributions in FixMin, RelMin, and ProgMin for the three player types.

Appendix B

EXPERIMENT INSTRUCTIONS (PROGMIN)

You participate in an economic decision experiment, in which you can earn money. How much each of you will earn depends on your personal decisions and those of other participants in the experiment. Each participant makes his decisions at a computer, isolated from the others. We ask you not to talk to other participants.

The experiment consists of 25 rounds. In the beginning of the experiment you will be randomly matched with three other persons to build a group of four. You will remain in this group during the entire experiment. You will not know the identity of your group members at any time, though.

Each group member is endowed in each round with a certain amount of tokens. Player 1 is endowed with 10 tokens per round. Players 2 and 3 are endowed with an amount of 15 tokens each. Player 4 is endowed with 20 tokens per round. The individual player numbers (and thus the individual endowments) will be randomly assigned and announced at the beginning of the experiment.

DECISIONS

In each of the 25 rounds, all group members each group member has to decide on how to allocate her/his tokens between two alternatives, called X and Y. The return of a token, in experimental currency units (ECU), is different for the two alternatives. The return of the allocation decision is determined as follows:

Each token that you contribute to X yields a return of 2 ECU. If you contribute nothing to X, your return from X is zero.

Each token that you contribute to Y, yields to you and to each of the other group members a return of 1 ECU. You may thus have a positive return from Y even if you yourself don't contribute anything to Y.

During the allocation of your tokens, you must note that you are required to contribute a minimum contribution to Y. This minimum contribution is 2 tokens for Player 1, 6 tokens for Players 2 and 3 each, and 10 tokens for Player 4.

Group Member	Endowment (tokens)	Mandatory Contribution to Y
Player 1	10	2
Player 2	15	6
Player 3	15	6
Player 4	20	10

You may allocate your tokens, above the minimum contribution to Y, to X or to Y only, but you may also allocate them among both alternatives. However, only entire tokens may be contributed. In the decision box on your screen you need to enter, for each alternative, the number of tokens that you want to allocate. If you do not want to contribute anything to X or Y, you need to type in a zero. The sum of the tokens contributed to X and Y must be always equal to your endowment. This means that the entire token endowment has to be allocated among X and Y. With the <Tab> key you can switch among the entry fields. The entries have to be confirmed by clicking on <OK>.

Your individual return per round is the sum of your returns from X and Y and is calculated as follows:

Return = $2 \times$ (your contribution to X) + (sum of tokens contributed to Y in your group).

PAYMENT

At the end of the experiment, you will be paid based on your individual total profit over all 25 rounds. Your individual total profit in ECU will be converted into \in (1 ECU = 0.01 \in) and paid to you in cash. You are paid at the end of the experiment. The payment is carried out individually and anonymously.

AVAILABLE INFORMATION

In each round, you will see an overview table on your screen which provides you with the results of all previous rounds that you have played. The results include the following information for each round:

Your endowment, your mandatory minimum contribution to Y, your contribution to X, your contribution to Y, the individual contributions to Y of each of the other group members, your return from X, your return from Y, your round profit, and your total profit.

If you want to see the results of earlier rounds, which are no longer visible in the table, please use the scroll function on the right side of the table.

We ask you now to go to the computer with your participation number. There you have to click on <Continue>. You then will be given on your screen a number of questions regarding these instructions. If you have any questions please address yourself to the experimenter. Only when all participants have correctly answered all questions, the experiment starts.